\begin{table}%t1-1
\caption{\label{tab:survey98}Observed and derived parameters
of surveyed sources.}
%
\par
%\centering
\par
\begin{tabular}{lllll rrccc rrccc}
\hline\hline 
{Source} &\multicolumn{6}{c}{Observed Parameters} & &\multicolumn{7}{c}{Derived Parameters}\\
\cline{2-7}\cline{9-15} \noalign{\smallskip}
{} & {$\alpha$(J2000)} & {$\delta$(J2000)} & {$T_{\rm mb12}$} & {$T_{\rm mb13}$} & {$V_{\rm LSR13}$} & {$\Delta V_{13}$} & & {$D$} & {$L_{\rm bol}$} & {$T_{\rm ex}$} & {$\tau_{13}$} & {$N(^{13}{\rm CO})$} & {$N({\rm H}_{2})$} & {Ref.}\\
 {\textit{IRAS} } & {(h m s)} & {($^{\circ}~^{\prime}~^{\prime\prime}$)} & {(K)} & {(K)} & {($\rm{km~s}^{-1}$)} & {($\rm{km~s}^{-1}$)} & & {(kpc)} & {(10$^{3}~L_{\odot}$)} & {(K)} & {} & {($10^{15}~{\rm cm}^{-2}$)} & {($10^{21}~{\rm cm}^{-2}$)} & {}\\
 {(1)}& {(2)} & {(3)} & {(4)} & {(5)} & {(6)} & {(7)} & & {(8)} & {(9)} & {(10)} & {(11)} & {(12)} & {(13)} & {(14)}\\
\hline
00117+6412$^{m}$ &00 14 27.7 &+64 28 46 & 5.72(0.12) & 2.55(0.05) & --36.06(0.01) &2.77(0.03) & &  3.6 & 4.29 & 10.50 &0.57 & 5.53 & 7.87 &1N, 2, 3N, 4N, 6, 7, 8     \\
00412+6638 &00 44 15.2 &+66 54 41 & 2.64(0.07) & 0.81(0.14) & --68.00(0.22) &3.32(0.22) & & 7.35 &15.25 &  6.96 &0.35 & 3.12 & 4.44 &2N, 4     \\
00468+6527 &00 49 55.8 &+65 43 39 & 2.31(0.08) & 1.17(0.07) & --63.67(0.03) &2.60(0.08) & & 6.74 &34.30 &  6.55 &0.67 & 4.51 & 6.42 &2, 4     \\
00557+5612$^{m}$ &00 58 44.4 &+56 28 16 & 8.37(0.05) & 3.72(0.05) & --29.96(0.01) &2.17(0.01) & & 2.94 & 1.47$^{u}$ & 13.34 &0.57 & 5.84 & 8.32 &11      \\
01134+6429 &01 16 48.1 &+64 45 37 & 5.03(0.06)$^{s}$ & 1.29(0.06) & --54.36(0.02) &2.20(0.06) & &    3$^a$ & 1.31 &  9.73 &0.29 & 2.18 & 3.10 &2, 4     \\
\\
02413+6037 &02 45 12.5 &+60 49 44 & 2.52(0.06)$^{s}$ & 0.93(0.04) & --61.56(0.02) &2.22(0.06) & & 7.47 &10.45$^{u}$ &  6.81 &0.44 & 2.58 & 3.68 &     \\
02485+6902 &02 53 07.2 &+69 14 36 & 4.11(0.04)$^{s}$ & 1.70(0.05)$^{s}$ & --10.52(0.01) &1.94(0.03) & & 1.09 & 0.21 &  8.70 &0.51 & 3.11 & 4.43 &2N     \\
02541+6208 &02 58 13.7 &+62 20 31 & 2.28(0.06) & 0.59(0.03) & --51.68(0.22) &2.09(0.22) & & 5.88 & 6.43$^{u}$ &  6.51 &0.29 & 1.55 & 2.20 &2, 4     \\
03101+5821$^{m}$ &03 14 04.7 &+58 33 08 & 4.84(0.05) & 2.37(0.04) & --38.27(0.01) &2.38(0.02) & &  4.2 & 1.20 &  9.52 &0.65 & 5.03 & 7.16 &2, 4N, 6     \\
03258+3104$^{m}$ &03 28 56.8 &+31 14 44 &13.57(0.11) & 6.78(0.05) &   7.56(0.01) &3.27(0.01) & & 0.22$^a$ & 0.03$^{u}$ & 18.76 &0.68 &15.29 &21.78 &1N     \\
\\
03414+3200$^{m}$ &03 44 36.4 &+32 09 25 &17.20(0.10) & 6.08(0.07) &   8.40(0.01) &1.71(0.01) & &  0.3$^b$ & 0.05$^{u}$ & 22.48 &0.43 & 7.44 &10.59 &1N, 2N     \\
04365+4717 &04 40 16.8 &+47 23 04 & 4.06(0.05)$^{s}$ & 1.65(0.08) & --34.75(0.03) &2.59(0.07) & &  6.9$^b$ & 4.37 &  8.64 &0.50 & 4.04 & 5.75 &2N, 4     \\
05155+0707 &05 18 17.1 &+07 11 01 &  ... & 3.34(0.06) &  --1.45(0.01) &2.37(0.02) & & 0.46$^a$ & 0.03 &   ... & ... &  ... &  ... &     \\
05168+3634$^{m}$ &05 20 16.2 &+36 37 21 & 3.39(0.06) & 2.61(0.09) & --15.18(0.22) &2.43(0.22) & & 6.08$^c$ &17.13 &  7.86 &1.37 &14.79 &21.05 &1N, 2N, 3, 5, 6, 7, 8, 10     \\
05221+4139 &05 25 39.8 &+41 41 50 & 2.16(0.06) & 0.79(0.07) & --25.81(0.05) &2.75(0.13) & &10.36 &32.75 &  6.36 &0.43 & 3.05 & 4.34 &2N     \\
\\
05271+3059 &05 30 21.2 &+31 01 27 & 2.32(0.07) & 0.75(0.06) & --19.73(0.22) &4.36(0.22) & & 16.5$^b$ &72.86 &  6.56 &0.37 & 4.22 & 6.01 &2     \\
05334+3149 &05 36 41.1 &+31 51 14 &  ... & 1.35(0.06) & --15.98(0.22) &3.77(0.22) & & 16.5$^b$ &97.73 &   ... & ... &  ... &  ... &2N, 3N     \\
05450+0019 &05 47 34.6 &+00 20 08 & 6.87(0.09) & 3.64(0.07) &   9.41(0.22) &4.19(0.22) & &  0.5$^d$ & 0.04 & 11.74 &0.73 &12.76 &18.17 &     \\
06067+2138$^{m}$ &06 09 48.0 &+21 38 11 &  ... & 3.49(0.05) &   1.73(0.22) &3.37(0.22) & & 0.61 & 0.03$^{u}$ &   ... & ... &  ... &  ... &2N, 6, 10     \\
06103+1523$^{m}$ &06 13 15.1 &+15 22 36 & 6.22(0.07) & 2.97(0.09) &  15.99(0.02) &3.01(0.04) & & 3.78 & 9.49 & 11.04 &0.63 & 6.91 & 9.84 &1N, 2N, 3N, 6, 7, 8     \\
\\
06104+1524A &06 13 21.3 &+15 23 57 &  ... & 3.24(0.19) &  16.26(0.03) &2.50(0.07) & & 3.87 &10.59 &   ... & ... &  ... &  ... &2N, 3N, 5N, 6     \\
06306+0232 &06 33 15.8 &+02 30 22 &  ... & 1.43(0.08) &  25.49(0.03) &2.28(0.06) & & 3.32 & 0.88 &   ... & ... &  ... &  ... &     \\
06331+1102 &06 35 56.0 &+11 00 18 &  ... & 2.66(0.10) &  22.71(0.22) &2.75(0.22) & & 3.89 & 2.58 &   ... & ... &  ... &  ... &6     \\
06381+1039 &06 40 58.0 &+10 36 49 & 9.47(0.19) & 2.64(0.10) &   7.64(0.02) &2.92(0.05) & & 1.11 & 0.20$^{u}$ & 14.50 &0.32 & 4.99 & 7.11 &6     \\
06382+1017 &06 41 03.3 &+10 15 01 & 7.70(0.08) & 3.52(0.09) &   7.60(0.01) &2.93(0.04) & &  1.1 & 0.17 & 12.63 &0.60 & 7.88 &11.22 &     \\
\\
06501+0143 &06 52 45.6 &+01 40 15 &  ... & 1.73(0.07) &  45.07(0.02) &2.85(0.05) & & 6.52 & 9.44 &  ... & ... &  ... &  ... &     \\
07024$-$1102$^{m}$ &07 04 45.7 &$-$11 07 15 &  ... & 4.47(0.06) &  16.90(0.22) &1.89(0.22) & & 1.64 & 0.57 &  ... & ... &  ... &  ... &6     \\
07111$-$1211 &07 13 29.9 &$-$12 16 51 &  ... & 1.22(0.08) &  15.71(0.03) &2.37(0.09) & & 1.53 & 0.20 &  ... & ... &  ... &  ... &     \\
07119$-$1210A &07 14 17.7 &$-$12 15 14 &  ... & 2.04(0.10) &  15.23(0.03) &2.67(0.06) & & 1.48 & 0.23 &  ... & ... &  ... &  ... &2N     \\
07207$-$1435 &07 23 01.3 &$-$14 41 33 &  ... & 1.95(0.13) &  53.17(0.04) &2.77(0.09) & & 5.58 &14.50 &  ... & ... &  ... &  ... &2       \\
\\
17576$-$1845 &18 00 34.3 &$-$18 45 17 &  ... &1.06(0.12) & 21.96(0.08) &4.38(0.20) & & 3.11 & 1.35 &  ... & ... &  ... &  ... &       \\
18145$-$1557 &18 17 26.7 &$-$15 56 20 &  ... &1.81(0.20) & 25.93(0.07) &4.09(0.19) & & 2.81 & 4.07 &  ... & ... &  ... &  ... &3N       \\
18205$-$1316 &18 23 21.6 &$-$13 15 02 &  ... &2.43(0.12) & 22.48(0.22) &2.89(0.22) & & 2.24 & 1.52 &  ... & ... &  ... &  ... &       \\
18236$-$1241 &18 26 24.7 &$-$12 39 37 &  ... &2.07(0.10)$^{s}$ & 64.57(0.03) &4.33(0.08) & &  4.6 &15.85 &  ... & ... &  ... &  ... &       \\
18278$-$0212 &18 30 28.0 &$-$02 10 48 &  ... &1.79(0.08) &  5.86(0.22) &3.01(0.22) & & 0.39 & 0.02 &  ... & ... &  ... &  ... &       \\
\hline
18301$-$0853 &18 32 55.2 &$-$08 51 23 &  ... &2.19(0.14) & 79.05(0.04) &3.35(0.10) & & 4.97 &13.66 &  ... & ... &  ... &  ... &       \\
18314$-$0820 &18 34 09.0 &$-$08 17 52 &  ... &4.07(0.08) &105.17(0.22) &4.01(0.22) & & 6.08 &24.52 &  ... & ... &  ... &  ... &3N       \\
18324$-$0855 &18 35 10.5 &$-$08 52 35 &  ... &0.92(0.10) &  3.72(0.03) &0.88(0.09) & & 0.26 & 0.03 &  ... & ... &  ... &  ... &1 \\
18358$-$0112 &18 38 25.9 &$-$01 09 51 &  ... &2.51(0.07) &  9.43(0.22) &3.67(0.22) & & 0.67 & 0.05 &  ... & ... &  ... &  ... &       \\
18403$-$0445 &18 38 34.4 &$-$03 32 06 &  ... &1.76(0.10)$^{s}$ & 45.74(0.03) &3.25(0.08) & & 3.16 & 4.66 &  ... & ... &  ... &  ... &1, 6       \\
\\
18502+0033 &18 52 47.6 &+00 36 51 &  ... &2.33(0.10) &104.90(0.03) &5.26(0.07) & &    1$^*$ & 0.50 &  ... & ... &  ... &  ... &       \\
18502+0034 &18 52 50.9 &+00 38 03 &  ... &2.40(0.12) &104.70(0.04) &5.55(0.09) & &    1$^*$ & 0.52 &  ... & ... &  ... &  ... &6       \\
18532+0047 &18 55 50.6 &+00 51 22 &  ... &1.24(0.09)$^{s}$ & 58.94(0.05) &4.65(0.13) & & 3.83 & 8.96 &  ... & ... &  ... &  ... &3N, 8       \\
18545+0202 &18 57 02.6 &+02 06 23 &  ... &1.45(0.10)$^{s}$ & 44.87(0.04) &3.68(0.13) & & 2.99 & 3.45 &  ... & ... &  ... &  ... &       \\
18578+0313 &19 00 21.1 &+03 17 45 &  ... &0.97(0.08)$^{s}$ & 59.74(0.06) &5.16(0.14) & & 3.93 & 3.53 &  ... & ... &  ... &  ... &       \\
\\
19002+0454 &19 02 42.0 &+04 58 49 &  ... &1.56(0.13)$^{s}$ & 68.58(0.06) &3.75(0.15) & & 4.67 & 8.71 &  ... & ... &  ... &  ... &3N, 7       \\
19011+0450 &19 03 36.9 &+04 55 15 &  ... &1.38(0.13) & 50.07(0.07) &4.52(0.17) & & 3.35 & 4.28 &  ... & ... &  ... &  ... &       \\
19029+0556 &19 05 23.8 &+06 01 24 &  ... &2.11(0.13) & 58.44(0.22) &4.54(0.22) & & 3.97 &12.65 &  ... & ... &  ... &  ... &       \\
19031+0621 &19 05 36.4 &+06 26 09 &  ... &1.65(0.10) & 73.36(0.03) &2.38(0.07) & & 5.41 &24.87 &  ... & ... &  ... &  ... &1       \\
19056+0624 &19 08 02.9 &+06 29 11 &  ... &1.16(0.12)$^{s}$ & 66.07(0.06) &3.44(0.17) & & 4.66 & 4.51 &  ... & ... &  ... &  ... &       \\
\\
19205+1358 &19 22 53.9 &+14 04 11 & 2.10(0.09) &0.60(0.07) & 61.86(0.07) &3.43(0.17) & &    1$^*$ & 0.31$^{u}$ & 6.28 &0.32 & 2.80 & 3.99 &       \\
19215+1410 &19 23 49.4 &+14 16 20 &  ... &0.80(0.07) & 58.00(0.07) &5.43(0.16) & &    1$^*$ & 0.16 &  ... & ... &  ... &  ... &6N       \\
19216+1658 &19 23 52.4 &+17 04 01 & 3.08(0.10) &0.92(0.07) &  1.67(0.06) &6.48(0.14) & & 10.6 &62.46$^{u}$ & 7.50 &0.34 & 6.18 & 8.80 &       \\
19282+1742 &19 30 30.5 &+17 48 30 &  ... &2.82(0.08) & 61.14(0.22) &2.84(0.22) & &    1$^*$ & 0.49 &  ... & ... &  ... &  ... &3N, 6       \\
19286+1722 &19 30 54.0 &+17 28 46 &  ... &0.88(0.07) & 45.48(0.05) &3.24(0.10) & & 4.78 & 6.57 &  ... & ... &  ... &  ... &6       \\
\\
19291+1713 &19 31 23.4 &+17 19 42 &  ... &1.45(0.07)$^{s}$ & 48.07(0.03) &3.41(0.07) & &    1$^*$ & 0.38 &  ... & ... &  ... &  ... &       \\
19298+1707 &19 32 08.5 &+17 13 35 & 4.97(0.15)$^{s}$ &0.87(0.07) & 56.61(0.05) &2.64(0.11) & &    1$^*$ & 0.19$^{u}$ & 9.67 &0.19 & 1.69 & 2.40 &       \\
19348+2229 &19 36 59.8 &+22 36 08 &  ... &0.82(0.05)$^{s}$ & 28.48(0.02) &1.31(0.05) & & 2.66 & 1.10 &  ... & ... &  ... &  ... &6N       \\
19368+2239 &19 38 58.1 &+22 46 32 & 6.19(0.08)$^{s}$ &3.20(0.09) & 36.35(0.22) &3.46(0.22) & &    1$^*$ & 0.33 &11.01 &0.70 & 9.47 &13.48 &3N, 6, 7, 8       \\
19406+2333 &19 42 44.4 &+23 40 30 & 3.03(0.05)$^{s}$ &1.06(0.08) &  1.01(0.05) &3.56(0.12) & & 8.74 & 8.87 & 7.44 &0.41 & 4.09 & 5.83 &6       \\
\\
19413+2349 &19 43 28.3 &+23 56 58 & 9.20(0.16) &3.85(0.09) & 22.33(0.01) &1.88(0.03) & & 2.02 & 0.60 &14.22 &0.53 & 5.19 & 7.39 &6       \\
19415+2312 &19 43 39.7 &+23 20 06 & 6.79(0.07) &2.05(0.06) & 27.53(0.02) &5.51(0.05) & & 2.72 & 2.03$^{u}$ &11.66 &0.35 & 7.97 &11.35 &6       \\
19560+3135 &19 58 03.3 &+31 44 07 & 4.11(0.09) &1.11(0.11) &--65.10(0.22) &2.19(0.22) & &13.29 &50.84 & 8.70 &0.30 & 2.08 & 2.96 &3N       \\
19589+3320 &20 00 52.6 &+33 29 08 & 7.34(0.10) &1.88(0.08) &--22.51(0.03) &4.78(0.07) & & 8.44 &37.18 &12.25 &0.29 & 6.03 & 8.58 &1       \\
20050+2720 &20 07 06.7 &+27 28 53 & 3.79(0.08)$^{s}$ &2.85(0.11) &  6.39(0.02) &4.05(0.06) & &  0.7$^a$ & 0.26 & 8.33 &1.31 &15.98 &22.75 &3, 7  \\
\\
20062+3550 &20 08 09.8 &+35 59 20 & 5.09(0.07) &2.37(0.17) &  0.99(0.22) &2.26(0.22) & & 5.24 &18.02 & 9.80 &0.61 & 4.74 & 6.74 &1, 3, 6, 7, 8  \\
20067+3415$^{m}$ &20 08 41.3 &+34 24 19 & 4.40(0.06) &2.64(0.06) & 13.68(0.02) &3.62(0.04) & & 2.36 & 1.14 & 9.03 &0.88 & 9.57 &13.63 &6  \\
20094+2744 &20 11 29.1 &+27 53 16 &  ... &2.91(0.14) & 12.10(0.03) &2.59(0.06) & & 1.11 & 0.12 &  ... & ... &  ... &  ... &  \\
20103+3633 &20 12 13.9 &+36 42 59 & 2.53(0.09) &1.70(0.24) &--36.23(0.07) &2.65(0.18) & & 8.91 &29.26 & 6.83 &1.04 & 7.32 &10.42 &3N  \\
20116+3605 &20 13 33.6 &+36 14 55 & 3.38(0.10)$^{s}$ &1.15(0.07)$^{s}$ &--53.07(0.03) &2.50(0.07) & &10.69 &72.70 & 7.85 &0.40 & 2.89 & 4.11 &  \\
\\
20145+3645 &20 16 27.5 &+36 54 58 & 2.70(0.11) &1.10(0.07) &--56.37(0.04) &3.54(0.12) & &10.86 &50.37 & 7.04 &0.50 & 4.76 & 6.78 &  \\
20149+3913$^{m}$ &20 16 42.6 &+39 23 15 & 6.91(0.08) &4.61(0.13) &  3.82(0.02) &2.97(0.04) & &1.7$^f$ & 0.30$^{u}$ &11.79 &1.06 &16.85 &24.00 &6, 9  \\
20178+3723 &20 19 43.1 &+37 33 13 & 6.03(0.09) &1.96(0.10) &  3.93(0.02) &1.84(0.05) & & 4.08 & 1.02$^{u}$ &10.83 &0.38 & 2.69 & 3.83 &  \\
20227+4154 &20 24 31.4 &+42 04 17 & 7.74(0.08) &3.18(0.12) &  5.54(0.02) &2.72(0.05) & &    2$^a$ & 2.74 &12.68 &0.52 & 6.37 & 9.06 &3, 7  \\
20231+3440 &20 25 16.0 &+34 50 06 & 3.60(0.07) &1.90(0.14) &  5.46(0.04) &2.70(0.09) & &    1$^a$ & 0.19 & 8.11 &0.72 & 5.72 & 8.14 &  \\
\hline
20261+3922 &20 27 58.8 &+39 32 07 & 2.63(0.08)$^{s}$ &1.19(0.19) &--54.99(0.09) &2.79(0.20) & &10.01 &32.70 & 6.95 &0.57 & 4.28 & 6.09 &  \\
20281+4038 &20 29 54.8 &+40 48 52 & 5.50(0.08)$^{s}$ &2.05(0.06) & --2.51(0.02) &2.36(0.04) & & 4.11 & 6.56 &10.25 &0.45 & 3.86 & 5.50 &  \\
20290+4052 &20 30 50.8 &+41 02 25 & 3.70(0.08)$^{s}$ &1.60(0.10) & --1.51(0.04) &3.15(0.10) & & 3.89 & 2.79 & 8.23 &0.54 & 5.12 & 7.28 &  \\
20326+3757 &20 34 33.0 &+38 08 02 & 9.05(0.11) &2.99(0.09) &  2.79(0.22) &5.81(0.22) & & 3.66 & 2.74$^{u}$ &14.06 &0.39 &11.72 &16.69 &  \\
20330+4109 &20 34 48.5 &+41 20 21 & 4.07(0.11)$^{s}$ &0.67(0.07)$^{s}$ &--30.82(0.05) &2.19(0.12) & & 7.09 &10.04$^{u}$ & 8.65 &0.17 & 1.19 & 1.69 &  \\
\\
20345+4024 &20 36 24.1 &+40 35 08 & 4.12(0.10) &0.82(0.09) &  1.31(0.06) &2.58(0.13) & & 3.34 & 1.59$^{u}$ & 8.71 &0.21 & 1.73 & 2.47 &  \\ 
20444+4629 &20 46 08.3 &+46 40 41 & 6.51(0.09) &3.65(0.07) & --3.71(0.01) &2.09(0.02) & & 2.91 & 3.98 &11.36 &0.80 & 6.67 & 9.50 &2N, 3N, 7  \\
20508+4825 &20 52 28.2 &+48 36 30 & 1.55(0.05)$^{s}$ &0.92(0.07) & --6.63(0.04) &2.58(0.11) & & 2.98 & 1.17$^{u}$ & 5.55 &0.84 & 5.27 & 7.51 &2N  \\
21025+4912 &21 04 15.4 &+49 24 25 & 1.32(0.04)$^{s}$ &0.74(0.06) &--73.44(0.05) &2.86(0.12) & &10.08 & 9.15$^{u}$ & 5.23 &0.76 & 5.27 & 7.50 &2  \\
21246+5512 &21 26 14.4 &+55 25 57 & 0.76(0.05) &0.49(0.06) &--69.33(0.05) &1.84(0.13) & & 8.72 &15.33 & 4.37 &0.93 & 4.14 & 5.90 &2N  \\
\\
21293+5535 &21 30 55.7 &+55 48 49 & 4.42(0.06) &1.17(0.06) &--71.13(0.03) &4.13(0.09) & & 8.85 & 8.25$^{u}$ & 9.05 &0.30 & 3.97 & 5.65 &2, 4  \\
21334+5039 &21 35 09.2 &+50 53 09 & 4.64(0.07) &2.36(0.07) &--45.05(0.22) &2.75(0.22) & &    5$^a$ &23.53 & 9.30 &0.68 & 6.21 & 8.84 &2  \\
21379+5106 &21 39 40.8 &+51 20 35 & 2.88(0.07) &1.39(0.07) &--42.27(0.22) &2.17(0.22) & &  5.9 & 5.18 & 7.25 &0.63 & 3.73 & 5.32 &2  \\
21391+5026 &21 40 57.3 &+50 39 53 & 2.56(0.07) &1.02(0.07) &--40.55(0.04) &2.50(0.09) & & 5.77 & 5.67 & 6.86 &0.48 & 3.22 & 4.58 &2  \\
21391+5802$^{m}$ &21 40 42.4 &+58 16 10 &  ... &5.80(0.07) &  0.68(0.01) &2.99(0.02) & & 0.75$^e$ & 0.26 &  ... & ... &  ... &  ... &1N, 2N, 3, 7, 8  \\
\\
21418+5403 &21 43 29.8 &+54 16 56 & 3.62(0.06) &0.86(0.06) &--60.07(0.04) &3.24(0.10) & & 7.51 &22.24 & 8.13 &0.26 & 2.51 & 3.58 &2N  \\
21519+5613 &21 53 39.2 &+56 27 46 & 4.26(0.06) &2.07(0.08) &--62.72(0.02) &3.11(0.05) & &  7.3$^a$ &15.75 & 8.87 &0.64 & 6.31 & 8.98 &2, 3, 4, 8  \\
22051+5848 &22 06 50.7 &+59 02 47 & 2.40(0.06) &1.51(0.07) & --1.77(0.02) &1.39(0.04) & & 0.77 & 0.08$^{u}$ & 6.66 &0.93 & 3.38 & 4.82 &2N, 4  \\
22198+6336$^{m}$ &22 21 27.6 &+63 51 42 &10.55(0.01) &5.39(0.07) &--11.10(0.22) &2.65(0.22) & & 1.67 & 1.53$^{u}$ &15.63 &0.70 &11.49 &16.36 &1N, 2N, 3, 7, 8  \\
22305+5803 &22 32 24.3 &+58 18 58 & 4.57(0.07) &1.80(0.06) &--52.41(0.02) &2.68(0.05) & & 5.94 &14.28 & 9.22 &0.48 & 4.25 & 6.05 &2, 4, 6, 8  \\
\\
22506+5944$^{m}$ &22 52 38.6 &+60 00 56 &11.01(0.07) &3.18(0.05) &--51.61(0.22) &2.43(0.22) & &  3.5$^d$ & 6.83 &16.11 &0.34 & 5.56 & 7.92 &1N, 2N, 3, 4N, 6, 7, 8, 10  \\
22539+5758 &22 56 00.0 &+58 14 46 & 7.32(0.06) &3.59(0.07) &--54.19(0.01) &2.97(0.03) & &  3.5$^d$ &10.68 &12.23 &0.66 & 8.47 &12.07 &2, 4, 5  \\
23011+6126 &23 03 13.1 &+61 42 26 & 5.12(0.07) &2.25(0.06) &--11.03(0.22) &1.93(0.22) & & 0.73$^d$ & 0.09 & 9.83 &0.56 & 3.75 & 5.34 &2, 4, 6  \\
\hline
\end{tabular}
\medskip
\par
Note. Column (1) is source name shown in order of ascending
\textit{IRAS} number. Columns (2) and (3) list J2000 equatorial
coordinates of each source. Column (4) lists the main beam
temperature of $^{12}$CO~$J= 2{-}1$ if available, and
Cols. (5)--(7) list the main beam temperature, local standard of
rest velocity and line width of $^{13}$CO~$J= 2{-}1$,
with $1\sigma$ rms level in parenthesis. Column (8) is kinematic
distance except for otherwise labeled. Column (9) lists the
bolometric luminosity with asterisk (*) if one (or two)
\textit{IRAS} band flux is upper limit. Column (10) is excitation
temperature. Column (11) is the opacity of ${}^{13}$CO. Columns
(12), (13) present average ${}^{13}$CO and H$_{2}$ column densities.
Column (14) lists former works if available.
\par
References. (1) 6.7~GHz methanol maser survey
\citep{2000AetAS..143..269S}; (2) 22~GHz water maser survey
\citep{1993AetAS...98..589W}; (3) 22~GHz water maser
\citep{1991AetA...246..249P}; (4) 1665/67~MHz OH maser survey
\citep{1993AetAS...98..589W}; (5) HCO$^+$~$J= 1{-}0$
survey \citep{1987MNRAS.228...43R}; (6) CS~$J= 2{-}1$
survey \citep{1996AetAS..115...81B}; (7)
NH$_3$~(\textit{J, K})~=~(1, 1) (2, 2) survey
\citep{1996AetA...308..573M}; (8) CO~$J= 2{-}1$ outflow
mapping \citep{2005ApJ...625..864Z, 2006ApJ...643..978K}; (9) CO,
CS mapping \citep{2004AJ....128.1716A}; (10) CO mapping
\citep{2008MNRAS.391..869G}; (11) CO mapping
\citep{2003NewA....8..191L}. N means
non-detection. \\\\
$^{a-f}${~distance from $^{a}$ \cite{2004AetA...426..503W},
$^{b}$ \cite{1993AetAS...98..589W}, $^{c}$
\cite{1996AetA...308..573M}, $^{d}$ \cite{1993AetAS...98...51H},
$^{e}$ \cite{1979AetA....75..345M},
$^{f}$ \cite{2007AetA...476.1243M}};
$^{*}${~kinematic distance unavailable, set as 1~kpc;}
$^{m}${~mapped source;}
$^{s}${~more than one velocity component, only the
strongest one listed;}
$^{u}${~upper limit.}
\end{table}